Author's response to reviews

Title: Identifying research priorities for health care priority setting: A collaborative effort between managers and researchers

Authors:

Neale Smith (neale.smith@ubc.ca)
Craig Mitton (craig.mitton@ubc.ca)
Stuart Peacock (speacock@bccrc.ca)
Evelyn Cornelissen (ecorneli@interchange.ubc.ca)
Stuart MacLeod (smacleod@cw.bc.ca)

Version: 2 Date: 27 November 2008

Author's response to reviews:

The editors have requested revisions to the manuscript, prior to peer-review: specifically, the inclusion of an explicit "Methods" section. We have made these changes and they are included in the revised draft.